Author's response to reviews

Title: Mucins in neoplastic spectrum of colorectal polyps: can they provide predictions?

Authors:

Mahsa Molaei (m_molaei@sbmu.ac.ir)
Babak Khoshkrood Mansoori (kmbabak@gmail.com)
Reza Mashayekhi (rzmash@gmail.com)
Mohsen Vahedi (mohsenvahedi540@gmail.com)
Mohamad Amin Pourhoseingholi (aminphg@gmail.com)
Seyed Reza Fatemi (nedafat20@yahoo.com)
Mohammad Reza Zali (nnzali@hotmail.com)

Version: 5 Date: 12 February 2010

Author's response to reviews:

The name of the corresponding author is changes from "Babak Khoshkrood Mansoori" to "Mahsa Molaei". All other parts of the last submission of this manuscript are approved by all authors.